Author's response to reviews

Title: Suicidal tendencies and attitude towards freedom to choose suicide among Lithuanian schoolchildren results from three cross-sectional studies in 1994, 1998, and 2002

Authors:

Nida Zemaitiene (nida@med.kmu.lt)
Apolinaras Zaborskis (socped@kmu.lt)

Version: 3 Date: 5 August 2005

Author's response to reviews:

We have formatted the manuscript according to the journal style as you have requested in the previous letter:

1. The keywords section has been removed from the manuscript.
2. The figure and table placement text has been removed from the manuscript.
3. The section of study limitations has been included under the discussion heading.
4. The legend sections in Table 2 and Table 4 have been removed.
5. The heading "Additional files" has been removed.

Therefore we did not change the figures to minimise white space around the image. In our point of view, the layout of the figures is quite good.